Author's response to reviews

Title: Therapeutic Hypothermia for Perinatal Asphyxial Encephalopathy in a Low Resource Hospital Setting in Equatorial Africa: Pilot Randomized Controlled Trial of Cranial Ultrasound and Neurodevelopmental Follow up at 18-22 months

Authors:

Nicola J Robertson (n.robertson@ucl.ac.uk)
Cornelia F Hagmann (corneliahagmann@london.com)
Dominique Acolet (d.acolet@imperial.ac.uk)
Elizabeth Allen (Elizabeth.Allen@lshtm.ac.uk)
Natasha Nyombi (nyombin@yahoo.com)
Diana Elbourne (Diana.Elbourne@lshtm.ac.uk)
Anthony Costello (cihdcostello@gmail.com)
Ian Jacobs (i.jacobs@ucl.ac.uk)
Margaret Nakakeeto (mnakakeeto@yahoo.co.uk)
Frances Cowan (f.cowan@imperial.ac.uk)

Version: 2 Date: 5 October 2010

Author's response to reviews:

Dear Editors,

Thank you for asking me to change the format of the paper - the initial results are now placed in the Results section. In addition I have removed Figure 3 as I have only verbal consent for these pictures of the infants and I think best not to publish. I hope that the other figures are acceptable to you for publication.

Yours sincerely

Nikki Robertson